Author’s response to reviews

Title: Outcome of coronary plaque burden: a 10-year follow-up of aggressive medical management

Authors:

Victor K. Goh (vkmgoh@yahoo.com)
Chu-pak Lau (cplau@netvigator.com)
Stefan Mohlenkamp (stefan.moehlenkamp@uk-essen.de)
John A. Rumberger (jrumberger@theplc.net)
Stephan Achenbach (Stephan.Achenbach@uk-erlangen.de)
Matthew J. Budoff (mbudoff@labiomed.org)

Version: 6  Date: 27 February 2010

Author’s response to reviews:

In this latest revision, some typographical errors were noted and corrected. The email address of the corresponding author was expanded to include all of the two mentioned in the list of email addresses of all co-authors.